Author’s response to reviews

Title: Uveitis-glaucoma-hyphema syndrome with sclera-fixed posterior-chamber two-haptic intraocular lens in a highly myopic eye: a case report

Authors:

Yu Du (joycedu624@163.com)
Xiangjia Zhu (zhuxiangjia1982@126.com)
Jin Yang (jin_er76@hotmail.com)
Yinglei Zhang (elaina0101@163.com)
Lei Cai (clyc0318@126.com)
Yi Lu (luyieent@163.com)

Version: 2 Date: 31 Dec 2019

Author’s response to reviews:

Dear Editor,

Thank you very much. We are really grateful for your kindness and our second chance to revise this case report.

As requested, we have cropped figure 1 and figure 4 to remove the dates, time and any initials that are included in these images.

With regard to the CARE checklist, thank you for your kind reminder. We do not wish to publish it, so according to your suggestion, we have removed it from the file inventory and included a statement within the manuscript that we adhered to CARE guidelines/ methodology at the end of the Declarations section. Also, it has been stated in the funding section that funding body had no role in the design of the study, collection, analysis, interpretation of data, or in writing the manuscript.

The uploaded manuscript is the final form for publication without any tracked changes or highlighting. Again, thank you very much for your affirmation of our work. We do hope the revised manuscript could meet all your requirements for publication. If there are any other problems remained to be addressed, please let us know. We will do our best to revise it again. Thank you.

Yi Lu